Author's response to reviews

Title: Real-time PCR detection of Human Herpesvirus 1-5 in patients lacking clinical signs of a viral CNS infection

Authors:

Birgitta Sundén (birgitta.sunden@gmail.com)
Marie C Larsson (larsson.c.marie@gmail.com)
Tina Falkeborn (tina.lundkvist@liu.se)
Jakob Paues (jakob.paues@lio.se)
Urban Forsum (urban.forsum@liu.se)
Magnus Lindh (magnus.lindh@microbio.gu.se)
Liselotte Ydrenius (Liselott.Ydrenius@lio.se)
Britt Åkerlind (britt.akerlind@lio.se)
Lena Serrander (lena.serrander@lio.se)

Version: 2 Date: 19 July 2011

Author's response to reviews:

Dear Sir/Madame,
We have checked the manuscript for "typos" and hope that there are none left.
We have also changed the title and made changes pointed out to us by the referees through the method-, result and discussion sections.
We hope that we have answered to all points in an adequate way.
Sincerely yours,
Lena Serrander